Author’s response to reviews

Title: Risk assessment of diesel exhaust and lung cancer: combining human and animal studies after adjustment for biases in epidemiological studies.

Authors:

Xanthi Pedeli (xpedeli@med.uoa.gr)
Gerard Hoek (g.hoek@uu.nl)
Klea Katsouyanni (kkatsouy@med.uoa.gr)

Version: 5 Date: 5 April 2011

Author’s response to reviews:

Dear Environmental Health Editorial Team,

We are submitting our revised manuscript entitled “Risk assessment of diesel exhaust and lung cancer: combining human and animal studies after adjustment for biases in epidemiological studies” by Pedeli, Hoek and Katsouyanni.

We have just put the title and content of Table 7 in the same page.

On behalf of the authors,

Klea Katsouyanni